Author's response to reviews

Title: Measuring patient safety culture in Dutch primary care: psychometric characteristics of the SCOPE-PC questionnaire

Authors:

Natasha J Verbakel (N.J.Verbakel@umcutrecht.nl)
Dorien LM Zwart (d.zwart@umcutrecht.nl)
Maaike Langelaan (m.langelaan@nivel.nl)
Theo JM Verheij (T.J.M.Verheij@umcutrecht.nl)
Cordula Wagner (c.wagner@nivel.nl)

Version: 2 Date: 23 July 2012

Author's response to reviews:

The sections ‘competing interests’, ‘authors’ contribution’ and ‘acknowledgements’ are added. The ethics statement was not necessary because the study did not involve human subjects.